Author's response to reviews

Title: Early recognition of heart failure in patients with diabetes type 2 in primary care. A prospective diagnostic efficiency study. (UHFO-DM2)

Authors:

Leandra JM Boonman-de Winter (l.boonman@shlbreda.nl)
Frans H Rutten (F.H.Rutten@umcutrecht.nl)
Maarten J Cramer (M.J.M.Cramer@umcutrecht.nl)
Anho H Liem (A.Liem@zzInd.nl)
Marcel J Landman (majlandman@planet.nl)
Henk F van Stel (H.vanStel@umcutrecht.nl)
G. Ardine de Wit (Ardine.de.wit@rivm.nl)
Guy EHM Rutten (G.E.H.M.Rutten@umcutrecht.nl)
Paulien AW van Hessen (P.vanhessen@shlbreda.nl)
Arno W Hoes (A.W.Hoes@umcutrecht.nl)

Version: 3 Date: 10 December 2009

Author's response to reviews: see over
Date
10 December 2009

Dear sir/madam,

We are pleased to send you our revised manuscript “Early recognition of heart failure in patients with diabetes type 2 in primary care.”

We addressed all the comments in the revised manuscript:

1) We included a citation for the figure and the table in the methods/design paragraph.
2) The abbreviations were formatted as a paragraph.
3) We removed the citation for additional file from figure legend and changed the layout of figure according to details for preparing files for submission and removed the title in the figure.
4) We removed the visible vertical lines from our table and changed the table, because there were some content errors.

We are looking forward to your reply.

On behalf of all authors,

Yours sincerely,

Leandra Boonman-Winter, MSc, Clinical Epidemiologist
SHL, Center for Diagnostic Support in Primary care, Bredaseweg 165, 4072 LA Etten-Leur, the Netherlands
Julius Center for Health Sciences and Primary Care, University Medical Center Utrecht, the Netherlands